Author's response to reviews

Title: New concept of scapholunate dissociation treatment and novel modification of Brunelli procedure - anatomical study

Authors:

Ahmed Y Elsaftawy (falastiny@interia.pl)
Jerzy Jablecki (jerzy.jablecki@interia.pl)
Tomasz Jurek (tomasz.jurek@umed.wroc.pl)
Adam Domanasiewicz (dariadam@tlen.pl)
Bohdan Gworys (bohdan.gworys@umed.wroc.pl)

Version:2 Date:30 January 2014

Author's response to reviews:

• The authors declare that they have no competing interests.
• The authors declare that this manuscript was not funded by any institution or foundation.
• The authors hereby declare that the concept of novel methods presented in the manuscript has been entirely made up by dr Ahmed Elsaftawy. After the methods was approved by prof. Jerzy Jablecki, we started to implement the study. Dr Bohdan Gworys and Dr Tomasz Jurek enabled us to use their laboratories in order to realize the biomechanical test on cadavers. The lab study was carried out with the participation of dr Adam Domanasiewicz. The manuscript was prepared and written solely by dr Ahmed Elsaftawy and was further accepted by the co-authors. . The research complies with the Declaration of Helsinki and has bee approved by the Commission of Bioethics at Wroclaw Medical University – reference number KB – 894/2012.